Author's response to reviews

Title: Association between gross motor function and postural control in sitting in children with cerebral palsy: A study of 139 children

Authors:

Sergio Montero Mendoza (smm1@um.es)
Antonia Gómez-Conesa (agomez@um.es)
Maria Dolores Hidalgo Montesinos (mdhidalg@um.es)

Version: 4 Date: 21 July 2015

Author's response to reviews: see over
Dear Editors,

Thanks for your revision. We have emphasized the changes we made in the manuscript by using bold text.

We have added the answers to the different comments from reviewers in the next paragraphs.

We are looking forward to hearing from you soon.

Maria Dolores Hidalgo Montesinos, PhD
AUTHOR QUERIES

Reviewer 1: Regina Harbourne

Minor essential revisions:

On p. 6, line 15, please reword this segment of your sentence: "in order to improve the monitoring of the frequency of the PC". Frequency of what? What is the PC?

Response from authors:

There was a mistake in the paragraph. The correct sentence is as follows: "in order to improve the monitoring of the frequency of the CP" (Cerebral Palsy).

Editor's comment:

- Please state whether informed consent was written or verbal.
- Please refer to the journal Author Instructions for the purpose of the Authors' Information section, currently titled Author Details (this section is not compulsory). Please use initials rather than full names in the Author Contributions section.
- The tables are currently additional files. Please bring them into the main manuscript (at the end of the paper). If you wish to leave any as additional files, please ensure they are referred to as such where they are cited in the text.
- Please add figure legends (after the references).
Response from authors:

On p. 7, line 12, we have specified that “written informed consent of parents and the school management were obtained in all cases.”

We have used the references about the journal Author Instructions for the purpose of the Authors' Information section, currently titled Author Details and the initials rather than full names in the Author Contributions section. We have placed the tables at the end of the manuscript and figure legends (after the references).